Author’s response to reviews

Title: Two cases of perforated corneal ulcers complicating rheumatoid arthritis treated successfully by biological therapy

Authors:

Sara BENCHERIFA (sara.bencherifa@gmail.com)

Bouchra Amine (amine_bouchra@yahoo.fr)

Imane El Binoune (imane.elbin@gmail.com)

Samira Rostom (rostomsamira2003@yahoo.fr)

Rachid Bahiri (bahirirachid@yahoo.fr)

Version: 2 Date: 25 Oct 2019

Author’s response to reviews:

The response to reviewers is included in the cover letter.